Author's response to reviews

Title: A prospective randomised trial comparing nasogastric with intravenous hydration in children with bronchiolitis (protocol) The comparative rehydration in bronchiolitis study (CRIB)

Authors:

Ed Oakley (Ed.oakley@southernhealth.org.au)
Franz E Babl (franz.babl@rch.org.au)
Jason Acworth (Jason_Acworth@health.qld.gov.au)
Meredith Borland (Meredith.Borland@health.wa.gov.au)
David Krieser (David.Krieser@wh.org.au)
Jocelyn Neutze (Jocelyn.Neutze@middlemore.co.nz)
Theane Theophilos (theane.theophilos@mcri.edu.au)
Susan Donath (Susan.donath@mcri.edu.au)
Mike South (mike.south@rch.org.au)
Andrew Davidson (andrew.davidson@rch.org.au)

Version: 4 Date: 21 April 2010

Author's response to reviews: see over
16 April 2010

Mick Aulakh, M.Sc.
Assistant Editor, BMC Medicine
BMC-series
BioMed Central
Floor 6,
236 Gray’s Inn Road
London, WC1X 8HL

Dear Mr Aulakh

Re: Protocol for the prospective randomised trial comparing nasogastric with intravenous hydration in children with bronchiolitis.
The comparative rehydration in bronchiolitis study (CRIB)
MS: 1582445646340298

Thank you for the reviews comments. I have incorporated all the requested changes. Below I outline the changes made to the manuscript.

REVIWER’S COMMENTS;
1. In the limitations section authors should also mention:
   - infant below 2 mo are excluded, and thus, results are not extrapolable to this group, particularly important and with severe course in bronchiolitis
   - why not using a clinical scale for the assessing
   - the exclusion of those patients above normal daily requirements

All these comments have been incorporated and the limitation section has been altered accordingly.

Yours sincerely

Ed Oakley
Chair Paediatric Research in Emergency Departments International Collaborative. (PREDICT)